Author's response to reviews

Title: Postoperative peri-axillary seroma following axillary artery cannulation for surgical treatment of acute type A aortic dissection

Authors:

EFSTRATIOS E APOSTOLAKIS (stratisapostolakis@yahoo.gr)
NIKOLAOS G BAIKOUSIS (ngbaik@yahoo.com)
KONSTANTINOS KATSANOS (baikoussisn@yahoo.com)
MENELAOS KARANIKOLAS (kmenelaos@yahoo.com)

Version: 2 Date: 21 May 2010

Author's response to reviews:

Cover letter – Journal of Cardiothoracic Surgery

Patras, May 21, 2010

To: the Editor-in-Chief
Journal of Cardiothoracic Surgery

Dear Sir or Madam,

We herewith submit a revised case report entitled “Postoperative peri-axillary seroma following axillary artery cannulation for surgical treatment of acute aortic dissection”, to be considered for publication in the Journal of Cardiothoracic Surgery.

The manuscript was prepared in accordance with the Journal of Cardiothoracic Surgery guide for authors. Author contribution, informed consent and competing interest are provided. Word count was removed from the manuscript.

Sincerely

N. G. Baikoussis MD